Author's response to reviews

Title: Party package travel. alcohol use and related problems in a holiday resort - a mixed methods study.

Authors:

- Morten Hesse (mh@crf.au.dk)
- Sebastien Tutenges (stu@crf.au.dk)
- Sanna Schliewe (sannacopenhagen@hotmail.com)
- Tine Reinholdt (tine_rein@hotmail.com)

Version: 4 Date: 7 October 2008

Author's response to reviews:

Dear editor,

We have changed the headings and table layout as requested.

Sébastien Tutenges and Morten Hesse are the corresponding authors. This is noted in the footnote with their e-mail addresses. We hope this is the appropriate way.

Sincerely
Morten

Sincerely, on behalf of all authors

Morten Hesse